Author's response to reviews

Title: Acute heroin intoxication in an infant chronically exposed to cocaine and heroin: a case report

Authors:

Xavier Joya (jjoya@imim.es)
Bibiana Fríguls (bfriguls@imim.es)
Marta Simó (msimo@hsjd.org)
Ester Civit (ecivit@imim.es)
Rafael de la Torre (rtorre@imim.es)
Antonio Palomeque (apalomeque@hsjd.org)
Oriol Vall (ovall@imas.imim.es)
Simona Pichini (simona.pichini@iss.it)
Oscar Garcia-Algar (90458@imas.imim.es)

Version: 6 Date: 7 December 2010

Author's response to reviews: see over
Title: Acute opioid intoxication in an infant chronically exposed to cocaine and heroin

Authors: Xavier Joya, Bibiana Fríguls, Marta Simó, Ester Civit, Rafael de la Torre, Antonio Palomeque, Oriol Vall, Simona Pichini, Oscar García-Algar

Version: 3

Date: 07 December 2010

Dear Editor:

Through the electronic submission system of the Journal, we have uploaded the reviewed manuscript entitled "ACUTE HEROIN INTOXICATION IN AN INFANT CHRONICALLY EXPOSED TO COCAINE AND HEROINE", which has been accepted by the Editorial Board of JMCR for the section Case Reports.

We appreciate very much your attention to this submission.

Sincerely yours,

Oscar García-Algar
Paediatrics Unit
Hospital Universitari del Mar
Passeig Marítim 25-29
E-08003 Barcelona, Spain
Tel.: +34-93-2483000, fax +34-93-2483245
E-mail: 90458@imas.imim.es

Comments: we have made all the changes suggested (major and minor revisions)

Major revisions

1. The abstract is structured as follow: Introduction, Case presentation, Conclusions.

2. Patient gender is stated in the Case Presentation of the manuscript.

Minor revisions

1. Authors' contribution is formatted into a paragraph.
2. The end of the manuscript is reformatted.

3. Vertical lines from the table has been removed.

4. Typography has been checked.